Title: Epidermal growth factor induces HCCR expression via PI3K/Akt/mTOR signaling in PANC-1 pancreatic cancer cells

Authors:

Zekuan Xu (xuzekuan@njmu.edu.cn)
Yi Zhang (zhyi7963@yahoo.com.cn)
Guoxin Zhang (guoxinz@njmu.edu.cn)

Version: 4 Date: 6 April 2010

Author’s response to reviews: see over
Dear editor:

Thank you very much for your letter and advice. We have revised the paper, following the instruction by the Editorial Committee when the paper is accepted, and the manuscript was resubmitted for your consideration. and I look forward to hearing from you soon.

Thanks very much for your attention to our paper.

Yours sincerely,

Guoxin Zhang and Yi Zhang

Major revisions:

>Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs)

The author name of Jin Woo Kim was replaced with Jin W Kim.

>Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

All of the affiliation were written in full as demanded.

>Conclusions: please add a Conclusions section following the Discussion. Alternately, you may change Discussion to Discussion and Conclusions.

The conclusion section was added.

>Additional files: please add a section titled "Additional files" at the end of the manuscript (after the tables) listing the following for each file: the title of the data, and a short description of the data. An example is given below:

The section titled "Additional files" which contained all of the figure and table description were added.

>Minor revisions

>Title page: remove running title. Funding statement should appear in the Acknowledgements only.

The running title was removed as required.

>Abstract: please change objective to Background and remove keywords.

>Introduction: rename Background.

>Materials and methods: rename Methods.
We have renamed the above three titles.

>Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
Thanks for your advice, the figures and tables were embedded within Word files and subsequently uploaded.

>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We have checked the typography errors of the manuscript.